Author's response to reviews

**Title:** Epidermal growth factor induces HCCR expression via PI3K/Akt/mTOR signaling in PANC-1 pancreatic cancer cells

**Authors:**

Zekuan Xu (xuzekuan@njmu.edu.cn)
Yi Zhang (zhiyi7963@yahoo.com.cn)
Guoxin Zhang (guoxinz@njmu.edu.cn)
Jiakai Jiang (jackjiang78@sina.com)
Yang Yang (yangyangdouble@hotmail.com)
Ruihua Shi (ruihuashi@126.com)
Bo Hao (bohao@126.com)
Zhihong Zhang (Zhangzhih2001@yahoo.com.cn)
Zuhu Huang (zhhuang@njmu.edu.cn)
Jin W Kim (jinwoo@catholic.ac.kr)

**Version:** 7  **Date:** 18 April 2010

**Author's response to reviews:** see over
Dear editor:

I should like to express my appreciation to you for suggesting how to improve our paper. We are sending the revised manuscript according to your requests.

Thanks very much for your attention to our paper.

Sincerely yours,

Guoxin Zhang and Yi Zhang

>Major revisions (we require the author to make these changes)

>Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match. Only 3 authors are listed in the submission system, while there are 10 listed in the manuscript.

I apologize for the delay in listing authors in the submission system, the other 7 authors were added.

>Email address: please include email addresses for all the authors on the title page, using the following format:

>AB: abcd@institution.ac.uk
>EF: efgh@generic.co.uk
>IJ: IJKL@corporation.com

Emails of all 10 authors were inserted in title page in the manuscript.

>Figure 2: Please reduce the volume of text presented in the figure legends to no more than 250 or it will compete with the image for space in the final publication.

The figure legend of Figure 2 was modified as your consideration.

>Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., rather than table 1, 2 etc, and include an additional files section in the manuscript.

Thanks very much for your suggestion, the additional files were removed,
and the tables were added following the references section.